Dear Dr. Lee,

First of all, thank you very much for your ongoing support of the World Journal of Gastroenterology and for contributing a manuscript to it.

We are very sorry to inform you that after rigorous peer review and evaluation by the editorial office, your manuscript does not reach the World Journal of Gastroenterology’s publishing standard with regards to the academic quality and, therefore, cannot be published in this journal.

A key criterion for accepting a manuscript is that the article must obtain a rating of AAA or AAB or ABB (A: Excellent; B: Very good) for academic content, by all of the three peer reviewers. If none of these ratings are attained, the manuscript will be rejected.

However, due to the interesting topical scope of your manuscript (Manuscript NO: 70306), we can offer acceptance for publication in the World Journal of Clinical Cases, another journal of the Baishideng Publishing Group (BPG), provided the following requirement is met: the manuscript must be revised according to the current reviewers’ comments that have been provided. If you are unwilling, your manuscript will be rejected. We recognize that you may be disappointed regarding the decision of rejection for the World Journal of Gastroenterology, but we encourage you to pursue the offer of publication in World Journal of Clinical Cases.

Please make your decision whether you are interested in our offer to publish your manuscript in World Journal of Clinical Cases within 3 days. Please login to the F6Publishing system at https://www.f6publishing.com by entering your registered E-mail and password. After clicking on the “Author Login” button, please click on the “Transfer to Another Journal” under the “Revisions” heading to find your manuscript. Clicking on the “Handle” button allows you to choose to either accept our offer to publish your manuscript in World Journal of Clinical Cases or reject the offer.

The World Journal of Clinical Cases is a high-quality, online, open-access, single-blind peer-reviewed journal published by the BPG.

The following peer-review report and comments from the Editorial Office (Science Editor, Editorial Office Director, and Company Editor-in-Chief) are provided for your reference.

1 Peer-review report

Reviewer #1: The article is very instructive. Intraperitoneal Paragonimus without lung involvement has been reported very rarely, and cases with a negative stool test and normal eosinophil counts may be misdiagnosed as tuberculous peritonitis. I agree with the author that preoperative serologic test for P. westmani may help patients recall history of crustacean consumption and prevent unnecessary surgery.

2 Editorial Office’s comments

1) Science Editor: Please state the exact date of admittance of the patient at the hospital. Language Quality: Grade B (Minor language polishing) Scientific Quality: Grade B (Very good)

2) Editorial Office Director: I recommend the manuscript to be published in the World Journal of Clinical Cases. Before final acceptance, uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”.
3) **Company Editor-in-Chief:** I recommend the manuscript to be published in the World Journal of Clinical Cases. Before final acceptance, uniform presentation should be used for figures showing the same or similar contents; for example, "Figure 1Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...".

Best regards,

Lian-Sheng Ma, Editorial Office Director, Company Editor-in-Chief, Editorial Office

**Baishideng Publishing Group Inc**

7041 Koll Center Parkway, Suite 160, Pleasanton, CA 94566, USA

**Telephone:** +1-925-399-1568

**E-mail:** ls.ma@wjgnet.com

**Help Desk:** [https://www.f6publishing.com/HelpDesk](https://www.f6publishing.com/HelpDesk)

**Online Submission:** [https://www.f6publishing.com/](https://www.f6publishing.com/)

**https://www.wjgnet.com/**

**About the Baishideng**

The Baishideng has 29 years of experience in editing and publishing medical journals. We currently publish 47 clinical medical journals covering a wide range of topics, 7 of which are indexed in the Science Citation Index Expanded (SCIE), and 23 in PubMed Central (PMC) and PubMed. The 2020 impact factors among these are: 5.742 for the World Journal of Gastroenterology; 5.326 for the World Journal of Stem Cells; 4.571 for the World Journal of Psychiatry; 3.763 for the World Journal of Diabetes; 3.393 for the World Journal of Gastrointestinal Oncology; 2.582 for the World Journal of Gastrointestinal Surgery; and 1.337 for the World Journal of Clinical Cases.

The Baishideng is an internationally recognized professional publishing company, and it is a member of CrossRef, Committee on Publication Ethics, Open Access Scholarly Publishers Association, and Association of Learned and Professional Society Publishers.

---

Dear editors

At first, I would like to thank you very much for reviewing my manuscript in detail.

I added the date of admission recommended by the scientific editor to the "outcome and follow-up" section.

Also revised the figure legend as recommended by the Editorial office director.
I appreciate you once again for allowing me to resubmit to this wonderful journal.

Thank you very much

Best regards,

Chang Min Lee

Gyeongsang National University Hospital